Author's response to reviews

Title: Periodontal disease and atherosclerosis from the viewpoint of the relationship between Community Periodontal Index of Treatment Needs (CPITN) and Brachial-Ankle Pulse Wave Velocity (baPWV).

Authors:

Koichi Miyaki (miyaki@sc.itc.keio.ac.jp)
Katsunori Masaki (masaki@keiomed.net)
Mariko Naito (mnaito@med.nagoya-u.ac.jp)
Toru Naito (naito@college.fdcnet.ac.jp)
Keika Hoshi (keikah@icqhc.or.jp)
Shugo Tohyama (shugotohyama@w4.dion.ne.jp)
Asako Hara (Asako2307@aol.com)
Takeo Nakayama (nakayama@pbh.med.kyoto-u.ac.jp)

Version: 2 Date: 18 November 2005

Author's response to reviews:

Dear editors,

We are pleased to submit the enclosed manuscript, "Periodontal disease and atherosclerosis from the viewpoint of the relationship between Community Periodontal Index of Treatment Needs and Brachial-Ankle Pulse Wave Velocity," as an original article for publication in BMC Public Health.

This manuscript is entirely original, and it is not under review elsewhere. All authors take responsibility for the contents of the manuscript, and satisfy the requirements for authorship. The corresponding author who will deal with the paper prepublication can be reached at:

Takeo Nakayama, M.D., Ph.D.
Department of Health Informatics, Kyoto University School of Public Health, Yoshida-Konoe-chou, Saky-ku, Kyoto, 606-8501, Japan
Tel: +81-75-7534488; Fax: +81-75-7534497
e-mail: nakayama@pbh.med.kyoto-u.ac.jp

Sincerely yours,
Takeo Nakayama